Author's response to reviews

Title: Participants' understanding of randomized controlled trial (RCT) through informed consent procedures in the RCT for breast cancer screening, J-START

Authors:
Masahiro Kikuya (kikuyam@med.tohoku.ac.jp)
Yoko N Shiono (narikawayoko@med.tohoku.ac.jp)

Version: 2 Date: 29 May 2014

Author's response to reviews: see over
May 29, 2014

Doug Altman,
Editor-in-Chief,
Trials, BioMed Central Journals

Dear Dr. Doug Altman,

We appreciate to be informed the formatting requests about our manuscript. The manuscript has been revised as follow.

1. We moved key words below our Abstract.
2. We included the trial registration number and date of registration for ‘J-START’ at the bottom of the Abstract.

We would be grateful if you could confirm the revised manuscript.

Respectfully,

Corresponding author:
Masahiro KIKUYA MD. Ph.D.,
Division of Molecular Epidemiology, Tohoku University Graduate School of Medicine
Address: 1-1, Seiryo-machi, Aoba-ku, Sendai, 980-8574, JAPAN
Phone number: +81-22-717-8104 / Fax number: +81-22-717-8106
Email: kikuyam@med.tohoku.ac.jp